Title: Hyponatreemic Hypertensive Syndrome (HHS) in Children Presenting as Malignant Hypertension: Case Report

Authors:

Dr Mehul P Dixit (mdixit@peds.arizona.edu)
John D Hughes (jhughes@email.arizona.edu)
Andreas Theodorou (aat@peds.arizona.edu)
Naznin M Dixit (ndixit@peds.arizona.edu)

Version: 2 Date: 4 Dec 2003

Spelling mistake and grammatical corrections only.